Title: Unexpected delayed recovery from longterm renal failure in a 12-year-old girl with severe diffuse proliferative lupus nephritis

Authors:

Sophia Ross (sophia.ross@uk-erlangen.de)
Kerstin Benz (kerstin.benz@uk-erlangen.de)
Katja Sauerstein (Katja.Sauerstein@uk-erlangen.de)
Kerstin Amann (Kerstin.Amann@uk-erlangen.de)
Jörg Dötsch (joerg.doetsch@uk-koeln.de)
Katalin Dittrich (katalin.dittrich@medizin.uni-leipzig.de)

Version: 2 Date: 13 February 2012

Author’s response to reviews:

Dear members of the editorial board,

we have carefully revised our manuscript according to your requirements. In particular, we have performed the following changes:

- Abstract
  We structured the abstract accordingly.

- Conclusions
  We included a separate Conclusions section after the Discussion.

- Consent
  We included a separate section under the heading of Consent.

- Competing Interests
  We inserted a declaration of competing interests at the end of the article.

- Authors’ Contributions
  We include an Authors’ Contributions section to appear after the Competing Interests section.

- Acknowledgements:
  We inserted an acknowledgement section.

Hoping that the paper is now suitable for assessment in BMC Nephrology,
I remain with best regards,
Kerstin Amann